Author’s response to reviews

Title: Biopsy-proven vancomycin-induced acute kidney injury: a case report and literature review

Authors:
ANRI SAWADA (anri-sawada@nms.ac.jp)
Kunio Kawanishi (kukawanishi@ucsd.edu)
Shohei Morikawa (morikawa.shohei@kameda.jp)
Toshihiro Nakano (t.nakano.dr@gmail.com)
Mio Kodama (mio.dama1113@hotmail.co.jp)
Mitihiro Mitobe (mitobe.michihiro@kameda.jp)
Sekiko Taneda (sekikosekiko@gmail.com)
Junki Koike (j2koike@marianna-u.ac.jp)
Mamiko Ohara (oharam-ty@umin.net)
Yoji Nagashima (nagashima.yoji@twmu.ac.jp)
Kosaku Nitta (knitta@twmu.ac.jp)
Takahiro Mochizuki (mochizuki.takahiro@kameda.jp)

Version: 1 Date: 06 Oct 2017

Author’s response to reviews:

Oct 6, 2017

Editor-in-Chief

BMC Nephrology
Dear Editor,

We thank for your comment.

In accordance with your comments, we have changed below.

1. We replace the statement of availability of data and materials subheading in our Declarations section.

2. We remove the age and sex characteristics from table 2.

We hope the revised version is now suitable for review and look forward to hearing from you at your earliest convenience.

Sincerely,

Anri Sawada

Department of Surgical Pathology, Tokyo Women's Medical University

8-1 Kawadacho, Shinjuku, Tokyo, 162-8666, Japan

Phone No: +81-3-3353-8111

Fax No: +81-3-5269-7321

Email Address: anri-sawada@nms.ac.jp